Author’s response to reviews

Title: Long-term exposure to air pollution and hospitalization for dementia

Authors:

Francesco Cerza (f.cerza@deplazio.it)
Matteo Renzi (m.renzi@deplazio.it)
Claudio Gariazzo (c.gariazzo@inail.it)
Marina Davoli (m.davoli@deplazio.it)
Paola Michelozzi (p.michelozzi@deplazio.it)
Francesco Forastiere (fran.forastiere@gmail.com)
Giulia Cesaroni (g.cesaroni@deplazio.it)

Version: 2 Date: 31 Jul 2019

Author’s response to reviews:

Reviewer #2: Authors have addressed all of my comments except including footnote about relevant national and/or EU air pollution standards to Table 3. In their response to reviewer comments, the authors state that they have included this information. However the version of Table 3 resubmitted does not have this footnote.

Thanks for the remark. We have added the footnote in table 3.